

## **Author's response to reviews**

**Title:** Analysing the role of complexity in explaining the fortunes of technology programmes: empirical application of the NASSS framework

### **Authors:**

Trisha Greenhalgh (trish.greenhalgh@phc.ox.ac.uk)

Joseph Wherton (joseph.wherton@phc.ox.ac.uk)

Chrysanthi Papoutsi (chrysanthi.papoutsi@phc.ox.ac.uk)

Jenni Lynch (j.lynch5@herts.ac.uk)

Gemma Hughes (gemma.hughes@gtc.ox.ac.uk)

Christine A'Court (christine.a'court@nhs.net)

Sue Hinder (suehinderrafresearch@hotmail.co.uk)

Rob Procter (rob.procter@warwick.ac.uk)

Sara Shaw (sara.shaw@phc.ox.ac.uk)

**Version: 1 Date:** 16 Jan 2018

### **Author's response to reviews:**

Thanks for returning it to us. We've shed 2000 words from the paper, which is now 6600 words long.

Trish